Author’s response to reviews

Title: Prevalence and factors associated with one-year mortality of infectious diseases among elderly emergency department patients in a middle-income country

Authors:

Maythita Ittisanyakorn (theseal13@gmail.com)
Sukkhum Ruchichanantakul (kuibma@gmail.com)
Alissara Vanichkulbodee (alissara115@gmail.com)
Jiraporn Sri-on (jiraporn.rew@gmail.com)

Version: 2 Date: 13 Jul 2019

Author’s response to reviews:

Thank you for the valuable comments.

We added in the declaration section Declarations

- Ethics approval and consent to participate

  The study was approve by Ethic Review Board of Faculty of Medicine, Vajira Hospital, Navamindradhiraj University. The COA number is 16/2560. The Faculty of medicine, Vajira hospital, Navamindradhiraj University IRB permissions were required to access the raw data.

- Consent for publication

  Not applicable

- Availability of data and materials

  Data can be obtained from the corresponding author upon reasonable re-quest.

- Competing of interests

  The authors declare no conflict of interests.

- Funding
This study had fully supported by Vajira Research Foundation Grant for Research Development. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

- Authors’ contributions

MI, AV, SR, and JS conceived and design a study. JS, MI acquisition of the data. JS, MI analyses and interpretation of the data. JS, and MI drafted of the manuscript, JS, MI, AV, SR, critical revision of the manuscript for important in-tellectual content and statistical expertise. All authors approved the final version of the manuscript to be published.